Author's response to reviews

Title: Isoflurane Preconditioning at Clinically Relevant Doses Induce Protective Effects of Heme oxygenase-1 on Hepatic Ischemia Reperfusion in Rats

Authors:

Xin Lv (xinlvg@hotmail.com)
Liqun Yang (lqyang72721@hotmail.com)
Yantao Liu (liuyangtao21896@163.com)
Weifeng Yu (ywf808@yeah.net)
Hao Lv (lhmousezhu@yahoo.com)
Feixiang Wu (Feixiangwu@yahoo.com.cn)

Version: 2 Date: 8 December 2010

Author's response to reviews:

Isoflurane Preconditioning at Clinically Relevant Doses Induce Protective Effects of Heme oxygenase-1 on Hepatic Ischemia Reperfusion in Rats

Xin Lv, Liqun Yang#, Kumming Tao, Yangtao Liu, Tian Yang, Guozhong, Chen$
Weifeng Yu* , Hao Lv, Feixiang Wu

Department of Anesthesia and Intensive Care, Eastern Hepatobiliary Surgery Hospital, the Second Military Medical University, 225 Changhai Road, Shanghai 200438, China

$ Department of Anesthesia, Fuzhou general hospital affiliated to Nan Jing Military District. Fuzhou, 350000, China

# Drs Lv and Yang contributed equally to this work.

Address correspondence to Weifeng Yu, MD, PhD, Department of Anesthesia and Intensive Care, Eastern Hepatobiliary Surgery Hospital, the Second Military Medical University, 225 Changhai Road, Shanghai 200438, China. Address e-mail to: ywf808@sohu.com

Adress co-correspondence to Guozhong Chen, MD, Department of Anesthesia, Fuzhou general hospital affiliated to Nan Jing Military District, Fuzhou, 350000, China. Address e-mail to: cgzssq2000@sina.com

Financial Support:
Supported by National Natural Science Foundation of China Grant No. 30700788 and Medical Key Program of Nan Jing Military District No.07Z039